\begin{table}%t1
\center%
\caption{\label{tab:experiments}Table~of the experiments which are used for the model.}
\begin{tabular}{lccccl}
    \hline\hline\noalign{\smallskip}
           &  Projectile mass    & Collision velocity & Micro-gravity  & Collisional outcome                            & Reference \\%
           &  $m_{\rm p}$ [g] & $v$ [\cms]  &  & (see Fig.~\ref{fig:pictograms}) & \\%
    \hline \noalign{\smallskip}
    Exp. 1  & $7.2\times 10^{-12}$--$7.2\times 10^{-9}$     & 0.1--1             & yes & S1         & \citet{BlumEtal:1998, BlumEtal:2002},\\
           &                                               &                      &     &            & \citet{WurmBlum:1998}\\
    Exp. 2  & $7.2\times 10^{-12}$--$2.0\times 10^{-10}$    & 10--50             & yes & S1         & \citet{WurmBlum:1998}\\
    Exp. 3  & $3.5\times 10^{-12}$--$3.5\times 10^{-10}$    & 0.02--0.17         & yes & S1         & \citet{BlumEtal:2000},\\
           & $1.0\times 10^{-12}$--$1.0\times 10^{-10}$    & 0.04--0.46         & yes & S1         & \citet{KrauseBlum:2004}\\
    Exp. 4  & $1.2\times 10^{-10}$--$4.3\times 10^{-10}$    & 7--$1~000$        & yes & S2         & \citet{BlumWurm:2000}\\
    Exp. 5  & $2\times 10^{-3}$--$7\times 10^{-3}$          & 15--390            & yes & B1, F1     & \citet{BlumMuench:1993}\\
           & $10^{-5}$-- $10^{-4}$                         & 15--390            & yes & B1, F1     & \\
    Exp. 6  & $10^{-6}$--$10^{-4}$                        & 10--170            & yes & S2, S3     & \citet{LangkowskiEtal:2008}\\
           & $10^{-4}$--$3\times 10^{-3}$                 & 50--200            & yes & B2, S2, S3 & \\
           & $2.5\times 10^{-5}$--$3\times 10^{-3}$        & 200--300           & yes & S3         & \\
    Exp. 7  & $10^{-3}$--$3\times 10^{-2}$                 & 20--300            & yes & S3         & \citet{BlumWurm:2008}\\
    Exp. 8  & $10^{-3}$--$3.2\times 10^{-2}$               & 16--89             & no  & S3         & \citet{GuettlerEtal:2009a}\\
    Exp. 9  & $10^{-3}$--$10^{-2}$                        & 10--40             & yes & B1         & \heisselmann\\
           & $10^{-3}$--$10^{-2}$                        & 5--20              & yes & B1         & \\
    Exp. 10 & $2\times 10^{-3}$--$5\times 10^{-3}$          & 1--30              & no  & B1         & \citet{WeidlingEtal:2009}\\
    Exp. 11 & $1.6\times 10^{-4}$--$3.4 \times 10^{-2}$     & 320--570           & yes & F1         & \citet{Lammel:2008}\\
    Exp. 12 & $3.5\times 10^{-15}$                           & $1~500$--$6~000$ & no  & F2         & \schraepler\\
    Exp. 13 & 0.2--0.3                                    & $1~650$--$3~750$ & no  & F2         & \citet{WurmEtal:2005a}\\
    Exp. 14 & 0.2--0.3                                    & 350--$2~150$      & yes & F2         & \citet{ParaskovEtal:2007}\\
    Exp. 15 & 0.39                                          & 600--$2~400$      & no  & S4         & \citet{WurmEtal:2005b}\\
    Exp. 16 & $4\times 10^{-7}$--$5\times 10^{-5}$          & 700--850           & no  & S4         & \citet{TeiserWurm:2009b}\\
    Exp. 17 & $1.6\times 10^{-4}$--$2.0\times 10^{-2}$      & 100--$1~000$      & no  & S4         & Sect.~\ref{sec:new_exp_1}\\
    Exp. 18 & $10^{-9}$--$10^{-4}$                        & 10--$1~000$       & no  & B1, S2, S4 & Sect.~\ref{sec:new_exp_2}\\
    Exp. 19 & $1.5\times 10^{-3}$--$3.2\times 10^{-3}$      & 200--700           & yes & S4, F3     & Sect.~\ref{sec:new_exp_3}\\
    \hline
\end{tabular}\vspace*{1.85mm}
\end{table}